Author's response to reviews

Title: Decision making on organ donation: the dilemmas of relatives of potential brain dead donors

Authors:

Jack de Groot (jack.degroot@radboudumc.nl)
Maria van Hoek (marline.vanhoek@radboudumc.nl)
Cornelia Hoedemaekers (astrid.hoedemaekers@radboudumc.nl)
Andries Hoitsma (andries.hoitsma@radboudumc.nl)
Wim Smeets (wim.smeets@radboudumc.nl)
Myrra Vernooij-Dassen (myrra.vernooij-dassen@radboudumc.nl)
Evert van Leeuwen (evert.vanleeuwen@radboudumc.nl)

Version: 2
Date: 17 December 2014

Author's response to reviews:

The integrative review was not published in 2011, but in 2012 (Transplantation. 2012;93(12):1196-211)

The article with experiences of requestors is published in 2014 (BMC Anesthesiology. 2014; 14:52; DOI: 10.1186/1471-2253-14-52)